Dear Editor,

The author has revised the article according to the opinions of the reviewers.

1. We have reformed introduction section in two paragraphs and have added the epidemiological literature data. In the second paragraph we stated clearly the aim of this paper.

2. We have separated the case presentation section in three different paragraphs.

3. We had reformed the discussion sections and added a paragraph discussing Mixed Müllerian tumors.

4. We invited a pathologist, but could not provide a clearer pathological picture due to the poor standard of pathological picture collection at that time.